\begin{thebibliography}{10}

\bibitem{aamich:vacuum}
Francis G.~Pease Albert A.~Michelson and Julius Pearson.
\newblock Measurement of the velocity of light in a partial vacuum.
\newblock {\em Astrophysics Journal}, 82:26--61, 1935.

\bibitem{Bradley:bio}
A.F.O'D. Alexander.
\newblock {J}ames {B}radley.
\newblock In C.C. Gillispie, editor, {\em Dictionary of {S}cientific
  {B}iography: {Vol II}}, pages 387--389. Charles Scribner's Sons, New York,
  1970.

\bibitem{Aristotle:Nicomachean}
Aristotle.
\newblock {Nicomachean Ethics}.
\newblock In Jonathan Barnes, editor, {\em {The Complete Works of Aristotle:
  The Revised Oxford Translation, Volume 2}}, pages 1729--1867. {Princeton
  University Press}, Princeton, 1984.
\newblock Translated by W.D. Ross with revision by J.O. Urmson.

\bibitem{Aristotle:heaven}
Aristotle.
\newblock {On the Heavens}.
\newblock In Jonathan Barnes, editor, {\em {The Complete Works of Aristotle:
  The Revised Oxford Translation, Volume 1}}, pages 447--511. {Princeton
  University Press}, Princeton, 1984.
\newblock Translated by J.L. Stocks.

\bibitem{Aristotle:soul}
Aristotle.
\newblock {On the Soul}.
\newblock In Jonathan Barnes, editor, {\em {The Complete Works of Aristotle:
  The Revised Oxford Translation, Volume 1}}, pages 641--692. {Princeton
  University Press}, Princeton, 1984.
\newblock Translated by J.A. Smith.

\bibitem{Aristotle:sense}
Aristotle.
\newblock {Sense and Sensibilia}.
\newblock In Jonathan Barnes, editor, {\em {The Complete Works of Aristotle:
  The Revised Oxford Translation, Volume 1}}, pages 693--713. {Princeton
  University Press}, Princeton, 1984.
\newblock Translated by J.I. Beare.

\bibitem{Bacon:Novum}
Francis Bacon.
\newblock Novum organum.
\newblock In Peter Urbach and John Gibson, editors, {\em Novum Organum: With
  Other Parts of the Great Instauration}, pages 33--293. Open Court, Chicago,
  1620.
\newblock 1994 translation.

\bibitem{Barnett:comparative}
Vic Barnett.
\newblock {\em Comparative Statistical Inference: Second Edition}.
\newblock John Wiley \& Sons, New York, 1982.

\bibitem{Beeckman:1629}
Isaac Beeckman.
\newblock {folio 340, 19 March 1629}.
\newblock In Christopher de~Waard, editor, {\em {Journal tenu par Isaac
  Beeckman de 1604 \`{a} 1634}}, volume iii, page 112. {Nijoff}, The Hague,
  1939-1953.

\bibitem{metre:def}
BIPM.
\newblock {\em Comptes Rendus des s\'{e}ances de la $17^e$ CGPM}.
\newblock BIPM, S\`{e}vres France, 1983.

\bibitem{Box:science}
G.E.P. Box.
\newblock Science and statistics.
\newblock {\em Journal of the American Statistical Association}, 71:791 -- 799,
  1976.

\bibitem{Chatfield:prob}
Chris Chatfield.
\newblock {\em Problem Solving. A Statistician's Guide (second edition)}.
\newblock Chapman and Hall, London, 1995.

\bibitem{cohen:1940}
I.B. Cohen.
\newblock Roemer and the first determination of the velocity of light.
\newblock {\em Isis}, 31:327 -- 379, 1940.

\bibitem{nauthist:1968}
Charles~H. Cotter.
\newblock {\em A History of Nautical Astronomy}.
\newblock American Elsevier, New York, 1968.

\bibitem{Deming:1953}
W.~Edwards Deming.
\newblock On the distinction between enumerative and analytic surveys.
\newblock {\em J. Amer Statist. Assoc.}, 48:244--255, 1953.

\bibitem{Drake:sci-bio}
Stillman Drake.
\newblock {\em Galileo at Work:His Scientific Biography}.
\newblock University of Chicago Press, Chicago, 1978.

\bibitem{Durbin:pres-rss}
J.~Durbin.
\newblock Statistics and statistical science.
\newblock {\em Journal of the Royal Statistical Society, A}, 150:pp. 177--191,
  1987.

\bibitem{edge:methods}
F.Y Edgeworth.
\newblock {\em Proceedings of the Silver Jubilee of the Royal Statistical
  Society}, 1884.

\bibitem{Feyerabend:method}
Paul Feyerabend.
\newblock {\em Against Method}.
\newblock Verso of New Left Books, New York, 1988.
\newblock (Revised Edition).

\bibitem{Fisher:methods}
R.A. Fisher.
\newblock {\em {Statistical Methods for Research Workers}}.
\newblock Oliver \$ Boyd, Edinburgh, 1925.
\newblock {Thirteenth edition, 1958}.

\bibitem{Galileo:starry}
Galileo Galilei.
\newblock {The Starry Messenger}.
\newblock In Stillman Drake, editor, {\em {Discoveries And Opinions of
  Galileo}}, pages 21--58. {Anchor Books, Doubleday}, New York, 1610.
\newblock Translated with an introduction and notes by Stillman Drake.

\bibitem{Galileo:duchess}
Galileo Galilei.
\newblock {Letter to Madame Christina of Lorraine Grand Duchess of Tuscany}
  concerning the use of {Biblical} quotations in matters of {Science}.
\newblock In Stillman Drake, editor, {\em {Discoveries And Opinions of
  Galileo}}, pages 173--216. {Anchor Books, Doubleday}, New York, 1615.
\newblock Translated with an introduction and notes by Stillman Drake.

\bibitem{Galileo:1638}
Galileo Galilei.
\newblock {\em Two New Sciences Including Centers of Gravity and Force of
  Percussion}.
\newblock The University of Wisconsin Press, Madison, Wisconsin, 1638.
\newblock Translated, with introduction and notes, by Stillman Drake, 1974.

\bibitem{Drake:disc}
Galileo Galilei.
\newblock {\em Discoveries and Opinions of {G}alileo}.
\newblock Anchor Books, Doubleday, New York, 1957.
\newblock Translated with an introduction and notes by Stillman Drake.

\bibitem{Gaukroger:Descartes}
Stephen Gaukroger.
\newblock {\em Descartes: An Intellectual Biography}.
\newblock Clarendon Press, Oxford, 1995.

\bibitem{Good:science}
I.J. Good.
\newblock {Scientific Method and Statistics}.
\newblock In {S. Kotz and N.L. Johnson}, editor, {\em {Encyclopedia of
  Statistical Sciences}}, volume~8, pages 291 -- 304. New York, 1988.

\bibitem{Hacking:phil}
Ian Hacking.
\newblock {\em Representing and Intervening: Introductory topics in the
  philosophy of natural science}.
\newblock Cambridge University Press, Cambridge, 1985.

\bibitem{HahnandMeeker:1993}
G.J. Hahn and W.Q Meeker.
\newblock Assumptions for statistical inference.
\newblock {\em The American Statistician}, 47:1--11, 1993.

\bibitem{Hand:decon}
David.~J. Hand.
\newblock Deconstructing statistical questions.
\newblock {\em J.R. Statist. Soc. A}, 157, 1994.

\bibitem{Hume:treatise}
David Hume.
\newblock {\em A Treatise on Human Nature}.
\newblock Oxford University Press, Oxford, 1739-1740.
\newblock {Analytical Index by L.A. Selby-Bigge, 1888; Second edition, 1978,
  with text revised and notes by P.H. Nidditch.}

\bibitem{kendall:vol1}
M.G. Kendall and A.~Stuart.
\newblock {\em The Advanced Theory of Statistics: Volume 1, Distribution
  Theory}.
\newblock Charles Griffin \& Company, London, 1958.
\newblock {First published in 1943, revised in 1958 with A. Stuart.}

\bibitem{Romer:bio}
Zdene\v{e}k Kobal.
\newblock {Ole Christensen R\"omer}.
\newblock In C.C. Gillispie, editor, {\em {Dictionary of Scientific Biography:
  Vol XI}}, pages 525--527. {Charles Scribner's Sons}, New York, 1975.

\bibitem{Kuhn:rev}
T.S. Kuhn.
\newblock {\em {The Structure of Scientific Revolutions}}.
\newblock Princeton University Press, Princeton, 1962.
\newblock {Second edition, 1970}.

\bibitem{Lakatos:meth}
Imre Lakatos.
\newblock {Falsification and the Methodology of Scientific Research
  Programmes}.
\newblock In John Worrall and Gregory Currie, editors, {\em {The methodology of
  scientific research programmes: Philosophical Papers Volume I}}, pages 8 --
  101. Cambridge University Press, Cambridge, 1978.
\newblock {First published in 1970}.

\bibitem{Lindberg:medieval}
David~C. Lindberg.
\newblock {Medieval Latin Theories of the Speed of Light}.
\newblock In Ren\'{e} Taton, editor, {\em {Roemer et la Vitesse de la
  Lumi\`{e}re}}, pages 45--72. Librairie Philosophique J. VRIN, Paris, 1978.

\bibitem{Losee:intro}
John Losee.
\newblock {\em {A historical introduction to the Philosophy of Science}}.
\newblock Oxford University Press, Oxford, 1993.
\newblock {Third Edition}.

\bibitem{Madden:methods}
Edward~H. Madden, editor.
\newblock {\em Theories of Scientific Method: The Renaissance through the
  Nineteenth Century}.
\newblock University of Washington Press, Seattle, 1966.

\bibitem{MichMorl:1887}
A.A. Michelson and E.W. Morley.
\newblock The motion of the earth relative to the luminiferous ether.
\newblock {\em American Journal of Science}, 34:333ff, 1887.

\bibitem{aamich:1880}
Albert~A. Michelson.
\newblock Experimental determination of the velocity of light made at the {U.S.
  Naval Academy, Annapolis}.
\newblock {\em Astronomical Papers}, 1:109--145, 1880.

\bibitem{MooreMcCabe:text}
David Moore and George McCabe.
\newblock {\em {Introduction to the Practice of Statistics}}.
\newblock W.H. Freeman and Company, New York, 1993.
\newblock {Second edition}.

\bibitem{Nelder:pres-rss}
J.A. Nelder.
\newblock Statistics, science and technology.
\newblock {\em Journal of the Royal Statistical Society, A}, 149:pp. 109--121,
  1986.

\bibitem{Newcomb:1882}
Simon Newcomb.
\newblock Measures of the velocity of light made under the direction of the
  {Secretary of the Navy} during the years 1880-'82.
\newblock {\em Astronomical Papers}, 3:107--239, 1882.

\bibitem{pearson:grammar}
Karl Pearson.
\newblock {\em {The Grammar of Science}}.
\newblock Meridian Books, New York, 1911.
\newblock {First published in 1892, revised 1900, final edition in 1911.
  Reprinted in 1957 with introduction by Ernest Nagel.}

\bibitem{Popper:logic}
Karl~R. Popper.
\newblock {\em The Logic of Scientific Discovery}.
\newblock Basic Books, New York, 1959.
\newblock {First English edition; original German edition published 1934.}

\bibitem{Descartes:speed}
Spyros Sakellariadis.
\newblock Descartes' experimental proof of the infinite velocity of light and
  huygens' rejoinder.
\newblock {\em Archive for History of Exact Sciences}, 26:1 -- 12, 1982.

\bibitem{sobel:long}
Dava Sobel.
\newblock {\em Longitude: The True Story of a Lone Genius Who Solved the
  Greatest Scientific Problem of His Time}.
\newblock Walker and Company, New York, 1995.

\bibitem{Stigler:robust}
Stephen~M. Stigler.
\newblock Do robust estimators work with {REAL} data? (with discussion).
\newblock {\em Annals of Statistics}, 5:1055--1098, 1977.

\bibitem{stig:hist}
Stephen~M. Stigler.
\newblock {\em The History of Statistics: The Measurement of Uncertainty before
  1900}.
\newblock Harvard University Press, Cambridge Massachusetts, 1986.

\bibitem{swenson:1972}
Loyd~S. {Swenson Jr.}
\newblock {\em The Ethereal Aether: A History of the {Michelson-Morley-Miller}
  Aether-Drift Experiments, 1880-1930}.
\newblock University of Texas Press, Austin Texas, 1972.

\bibitem{Cassini:bio}
Ren\'{e} Taton.
\newblock {Gian Domenico Cassini}.
\newblock In C.C. Gillispie, editor, {\em {Dictionary of Scientific Biography:
  Vol III}}, pages 100--104. Charles Scribner's Sons, New York, 1971.

\bibitem{Tukey:both}
J.W. Tukey.
\newblock We need both exploratory and confirmatory.
\newblock {\em The American Statistician}, 34:23--25, 1980.
\newblock {Reproduced in Volume IV of the Collected Works of John Tukey.}

\bibitem{Wheeler:meas}
Donald~J. Wheeler and Richard~W. Lyday.
\newblock {\em Evaluating the Measurement Process}.
\newblock SPC Press Inc., Knoxville, TN, 1989.

\bibitem{Youden:meas}
W.J. Youden.
\newblock {\em Experimentation and Measurement}.
\newblock National Science Teachers' Association, Washington, 1985.

\end{thebibliography}

\subsection{The Problem}
The purpose of the Problem stage is to provide a clear statement of what is to be
learned from the study. 
To do so, it is important to translate the contextual problem under study into
a language that can guide the design and implementation of the subsequent stages
of Statistical Method. Understanding what is to be learned from the study is so
important that it is surprising that it is rarely treated in any introduction to 
Statistical Method. In a cursory review, we could find no elementary text that provided a structure to
understand the  problem. 
For example, the popular text  by Moore and McCabe  \cite{MooreMcCabe:text} 
makes no mention of the role of Statistics in problem formulation. The cost of this 
omission is incalculable.

Two exceptions are the paper by Hand \cite{Hand:decon} and  Chatfield's book, \cite{Chatfield:prob}. Hand's 
aim was ``to stimulate debate about the need to formulate research questions sufficiently precisely
that they may be unambiguously and correctly matched  with statistical techniques''. He suggests five 
principles to aid in this matching but no structure or language. Chatfield provides excellent advice to get a 
clear understanding of the physical background to the situation under study,  to clarify the objectives
and to formulate the problem in statistical terms. One of our goals is to provide appropriate statistical terms, 
a language to formulate the problem, and a structured approach  to this formulation. 

We break the Problem stage into several sub-stages.
\subsubsection{Specifying the Target Units and Population}
Statistical method is concerned with gaining knowledge about a population. Here we borrow the
language of sample surveys \cite{cochran:1977} with the intent to apply it
broadly. The {\em
target
population} is the collection of {\em units} which we want to learn
about or which we want to apply the conclusions to.

To illustrate the terminology, in 1879, Michelson was keen to determine the speed of white light as
 it travels
between any two relatively stationary points in a vacuum.  A unit is one
transmission of such light between a source and destination, both located in a
vacuum. The target population is all such transmissions, before, during and
after 1879.

-poorly specified, looking back, sometimes
arbitrary
-finite or not philosophical - limits may be unknown, conceptually
uncountable in this case - does'nt matter!:wq

-one or two
-unit changes eg blood pressure over tieme, meas system study

draw a
conclusion.  A {\em unit} is
{\em Units and Target Population}-these specify the collective to which we are
interested in applying our learning.
\item
{\em Variates} - numerical or categorical values attached to every unit in the
target population (values of variates may differ from unit to unit). 
\item
{\em Population Attributes} - functions that apply to the entire 
population calculated through the variate values on individual units.
\item
{\em Problem Aspect}  - either {\em causative}, {\em predictive} or
{\em descriptive}. A problem with a causative aspect corresponds to one where
interest lies in investigating the nature of a causative relationship
between two or more variates in the target population. A problem has a predicitve aspect if the object is to 
predict the values of variates on one or more units in the target population. A problem has a descriptive 
aspect
if the object is to estimate or describe one or more attributes of the population. 
\end{enumerate}

To illustrate the terminology, in 1879, Michelson was keen to determine the speed of white light as it travels
between any two relatively stationary points in a vacuum.  A unit is one
transmission of such light between a source and destination, both located in a
vacuum. The target population is all such transmissions, before, during and
after 1879. The primary variate of interest, which we call the {\em response
variate}, is the speed of light associated with each such transmission.  There are
many other variates, which we call {\em explanatory variates}
attached to  each unit  such as the distance between the two
points, the motion of the points with respect to each other, properties of the
source and so on.  In Michelson's problem, there is no direct interest in these other
variates. 

The attribute of interest is the speed of light  averaged across all units in the target
population. This example is unusual in that it was believed that there is no
variation in the value of the response variate from unit to unit i.e that the speed of white light in a vacuum 
is constant.

The problem here is descriptive. The aim is to estimate a population attribute.
If Michelson had been attempting to show that the speed of light can be changed
by, for example, having the source move towards the destination, then the Problem
has a causative aspect. It is important to decide the aspect at the problem stage
because of the special requirements of the Plan needed to establish causation.

This language can be used to clarify many statistical issues. Consider, for example, Demings notion of an 
enumerative versus analytic study, \cite{Deming:1953}, \cite{HahnandMeeker:1993}.  If the target 
population can be listed so that a probabilistic sampling protocol giving every unit a 
positive inclusion probability can be used, the study is enumerative. Deming was particularly interested in 
industrial processes \footnote{We have found it is often useful to specify the target population by describing 
the process that generates the units. This process is called, naturally, the target process}, where the target 
population included units not yet produced. All studies on such processes are analytic. 

In applying Exploratory Data Analysis, the target population is the sample, the units that are included in the 
study. The object is to find interesting relationships among the usually many variates available. The 
relationships are defined as attributes in the target population. 

A clear specification of the attribute of interest can resolve many issues. Lord's paradox, as presented by 
\cite{Hand:decon}, is easily resolved by noting that it involves two different attributes. See our discussion to 
Hand. 

